RESEARCH

Open Access

Autologous umbilical cord blood mononuclear cell therapy for hypoplastic left heart syndrome: a nonrandomized control trial of the efficacy and safety of intramyocardial injections



Carlos Gallego-Navarro^{1,2}, James Jaggers³, Harold M. Burkhart⁴, Waldemar F. Carlo⁵, David L. Morales⁶, M. Yasir Qureshi^{2,7}, Joseph W. Rossano⁸, Clinton E. Hagen⁹, Drew K. Seisler², Susana Cantero Peral^{1,2} and Timothy J. Nelson^{1,2,7,9,10*}

Abstract

Background Preliminary phase I clinical trial results revealed that autologous umbilical cord blood-derived mononuclear cells (UCB-MNCs) preserved right ventricular cardiac function. To establish the efficacy of intramyocardial injections of an autologous UCB-MNC product at the time of stage II palliation surgery in patients with hypoplastic left heart syndrome (HLHS).

Methods A phase IIb, multicenter, open-label, nonrandomized study was conducted. Ninety-five children (fifty treated and forty-five controls) with HLHS and its variants, a history of stage I palliation surgery, and planned stage II palliation surgery at less than thirteen months were enrolled. We assessed coprimary efficacy endpoints for changes in right ventricular cardiac function through fractional area changes and longitudinal and circumferential strain, both in the short term (three months) and long term (twelve months). Second, we assessed changes in biomarkers of cardiac injury. Safety endpoints included severe adverse events (SAEs), changes in overall health through vital signs, and cumulative hospitalization.

Results Assessment of our coprimary efficacy endpoints revealed an unfavorable change in longitudinal cardiac strain in the treatment group compared with an improvement in strain in the control group (unadjusted p=.032) in the short term. No differences were observed between the groups in terms of other coprimary efficacy endpoints in the short or long term. A secondary assessment of biomarkers of cardiac injury revealed higher troponin T levels in the treatment group at three and six hours postsurgery. Regarding safety, no deaths related to the administered product or delivery procedure were reported. The treatment group presented a greater incidence (20%) of at least

*Correspondence: Timothy J. Nelson Nelson.Timothy@mayo.edu

Full list of author information is available at the end of the article



© The Author(s) 2025. **Open Access** This article is licensed under a Creative Commons Attribution-NonCommercial-NoDerivatives 4.0 International License, which permits any non-commercial use, sharing, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if you modified the licensed material. You do not have permission under this licence to share adapted material derived from this article or parts of it. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by-nc-nd/4.0/.

one SAE than the control group at three months (p =.048). Additionally, no statistically significant differences were found for the other safety endpoints.

Conclusion Intramyocardial injections of autologous UCB-MNC products into the right ventricular myocardium during stage II palliation surgery failed to enhance cardiac function in patients with hypoplastic left heart syndrome.

Registered on ClinicalTrials.gov Registered on ClinicalTrials.gov (NCT03779711) on 12/04/2018; URL: https://clinicaltrials.gov/ct2/show/NCT0377971.

Novelty and significance

What is known?

• To date, no prior efficacy studies have examined the use of autologous umbilical cord blood-derived mononuclear cells (UCB-MNCs) in patients with hypoplastic left heart syndrome following stage palliation surgeries.

• A preliminary phase I clinical trial noted that early follow-up (six months) after the administration of autologous UCB-MNC enhanced right ventricular cardiac function relative to that of controls, alongside greater weight percentiles [19].

What new information does this article contribute?

• Largest worldwide stem cell phase IIb clinical trial on congenital heart disease, specifically investigating the efficacy and safety of an autologous UCB-MNC product in patients with hypoplastic left heart syndrome.

• The autologous UCB-MNC product depicted a favorable safety profile, with no severe adverse events or fatalities related to the product or product administration.

• Despite the favorable safety profile observed, intramyocardial injections of the autologous UCB-MNC product did not yield improvements in cardiac function in the short or long term.

The preservation of right ventricular cardiac function remains a critical challenge in HLHS, as evidenced by several studies documenting a decline in right ventricular contractility over time at various stages of surgical palliation. The use of stem cells is effective in regenerating, remodeling, and renewing injured myocardium. Although preliminary clinical trials reported improved cardiac function following the administration of autologous UCB-MNC in HLHS patients, this study failed to demonstrate the preservation of cardiac function while maintaining safety. The multicenter nature and magnitude of this dataset are substantial and should not be understated; however, we must also acknowledge the limitations of our study, underscoring the need for further research to address them and optimize the therapeutic potential of UCB-MNC therapy for HLHS. We are committed to continuing our efforts to provide more bases and insights into the mechanisms and a better understanding of regenerative medicine to find solutions for congenital heart disease.

Keywords Hypoplastic left heart syndrome, Single ventricle, Umbilical cord blood stem cell transplantation, Mononuclear cells, Stem cell, Regenerative therapy

Background

Regenerative medicine emerged in the early 21st century as a promising approach to address the challenges faced by conventional medicine in effectively treating diseases and conditions that have shown disheartening outcomes. It has expanded the knowledge and provided a new perspective on the mechanisms of these diseases. This field encompasses a thorough understanding of stem cell biology and other basic sciences, such as biochemistry, genetics, immunology, and computer sciences. Its primary goal is to restore the functionality of weakened or damaged cells through a comprehensive process that involves manipulating healthy progenitor cells or undifferentiated stem cells, thereby facilitating the restoration and optimization of their inherent capabilities.

Congenital heart disease (CHD) has a prevalence of 8.1 per 1,000 live births [1]. HLHS, the most common form of single-ventricle CHD, has a prevalence of 2-3 per 10,000 live births and the highest mortality when

treatment is not offered in a timely manner at birth [1-4]. HLHS is associated with impaired heart function due to the underdevelopment of left heart structures (5). The management of this condition comprises multistage palliative surgeries, which start at birth, continue at 4–6 months of age, and end with Fontan surgery at approximately 18–48 months of age [6, 7]. Most newborns currently undergoing surgery for HLHS survive, with approximately two-thirds alive after stage III palliation; within this group, approximately 80% continue to survive for up to 20 years after that [8]. Despite improving survival rates, these children still face long-term risks attributed to multiorgan system failure, partly due to right ventricular dysfunction, resulting in an 8% transplant risk within 20 years [9, 10].

In recent years, numerous clinical trials have emerged as alternative approaches to improve cardiac function in children with CHD, particularly those with right ventricle physiology. Clinical trials involving CHD in children involve several factors to consider, including donor selection (allogeneic vs. autologous), the type of tissue (cardiac, bone marrow, or umbilical cord blood), the administration route (intracoronary, intramyocardial, epicardial, or intravenous), and the timing of administration during multistage surgical palliation [11].

In 2015, a team from Okayama University in Japan published the first phase I and later phase II clinical trials investigating transcoronary infusion of cardiac progenitor cells in patients with single-ventricle physiology [12]. Other trials include phase I/IIb clinical trials in which allogeneic human mesenchymal stem cell injections were used during stage I palliation surgery in patients with hypoplastic left hearts [13].

However, it is essential to acknowledge that the development of stem-cell clinical trials for CHD in children is years behind compared with the extensive experience from numerous trials in adult heart disease [14–17]. Our team focused on stem cell therapies involving autologous bone marrow and umbilical cord blood-derived mononuclear cells (UCB-MNCs). In 2019, we published the first groundbreaking phase I clinical trial involving single-dose intramyocardial injections of autologous UCB-MNC products in ten children with HLHS during the bidirectional Glenn shunt procedure (stage II palliation surgery). This trial successfully demonstrated the safety and feasibility of the autologous UCB-MNC product. In addition, the preservation of baseline right ventricular cardiac function and the normalized growth rates in the follow-up period motivated us to substantiate this therapy's evidence and benefits further [18]. Hence, we conducted this longitudinal phase IIb clinical trial to assess the safety and effectiveness of the autologous UCB-MNC product through intramyocardial injections in a group of patients with HLHS compared with control subjects.

Methods

Study design and participants

This study is a multicenter, prospective, open-label, nonrandomized, observational phase IIb clinical trial designed to determine the safety and efficacy of a UCB-MNC product delivered into the right ventricular myocardium of patients with HLHS and its variants at the time of stage II palliation surgery (Fig. 1). This study focused primarily on short- and long-term right ventricular cardiac function. Second, cardiac injury was measured through cardiac biomarkers [troponin T and B-type natriuretic peptide (BNP)], severe adverse events (SAEs), cumulative days of hospitalization, vital signs, and parents' perceptions of their child's overall health and developmental ability. This study was conducted at Mayo Clinic (Rochester, MN), Children's Hospital Colorado (Aurora, CO), Children's Hospital of Alabama (Birmingham, AL), Cincinnati Children's Hospital Medical Center (Cincinnati, OH), Ochsner Medical Center Jefferson (Jefferson, LA), Children's Hospital Los Angeles (Los Angeles, CA), Children's Hospital and Clinics of Minnesota (Minneapolis, MN), Oklahoma Children's Hospital, University of Oklahoma (Oklahoma City, OK), and Children's Hospital of Philadelphia (Philadelphia, PA).

Eligibility and enrollment

Before enrollment, expecting parent(s) with a prenatal diagnosis of HLHS or HLHS variants by fetal echocardiogram were enrolled in the ongoing Mayo Clinic umbilical cord blood (UCB) study for collection of UCB cells for HLHS (ClinicalTrials.gov identifier NCT01856049)



Fig. 1 Consort flow diagram showing the number of enrolled subjects, subject allocation into treatment groups, and primary endpoint data available

under an institutional review board-approved protocol. Our study population comprised fifty subjects enrolled in the treatment group and forty-five controls. Both groups met identical study eligibility criteria, study tests, and procedures, except for the requirements related to the acceptability of the UCB-MNC product and the sensitivity of the dimethyl sulfoxide (DMSO) reaction (treatment group only). The inclusion criteria included a diagnosis of HLHS or an HLHS variant with single right ventricular-dependent CHD, a history of prior stage I palliation surgery, being scheduled for stage II palliation surgery within less than thirteen months of age after stage I surgery, and having participated in the UCB collection protocol with the autologous UCB-MNC product that was deemed acceptable for clinical use (treatment group only). The exclusion criteria included a history of DMSO, parent(s) and legal guardian(s) unwilling to have their child participate, severe chronic diseases, extensive extracardiac syndromic features, a known history of cancer, and any of the following: complications of congenital heart disease, any condition requiring urgent or unplanned interventional procedures within 15 days before stage II palliation surgery.

The study was approved by the Institutional Review Boards (IRBs) of the corresponding site institutions, which are as follows: Colorado Multiple IRB; Children's Hospital Los Angeles IRB; Children's Minnesota IRB; The Children's Hospital of Philadelphia Research Institute IRB; Cincinnati Children's Hospital IRB; Ochsner Clinic Foundation IRB; The University of Oklahoma - IRB for the Protection of Human Subjects; The University of Alabama at Birmingham - Office of the IRB for Human Written informed consent was obtained from all parents and legal guardian(s) of the subjects meeting the eligibility criteria. After providing consent, a screening phase was conducted that included a preoperative baseline evaluation: clinical evaluation, laboratory and imaging tests, and questionnaires, followed by stage II palliation surgery, which included enrollment. Subjects were actively followed up after surgery via in-clinic visits and telephone follow-ups. The protocol utilized was monitored by both the Food and Drug Administration (FDA) and the IRBs. One of our authors had full access to all the data in the study and took responsibility for its integrity and data analysis.

Manufacturing, storage, transportation, and administration of the UCB-MNC product

The autologous UCB used to manufacture the cell-based product was collected under the UCB collection study protocol at birth and shipped to ReGen Theranostics, Inc. (Rochester, MN) for manufacturing. Manufacturing this cardiac-specific product requires a unique UCB collection and cell preparation that is different from the traditional UCB collection and processing protocols that allow direct thawing and delivery at surgery. The sponsor was responsible for reviewing the release criteria on the certificate of analysis from the manufacturer for each autologous product and determining the acceptability of the product's release to the investigational site for clinical use. During stage II palliation surgery, the frozen autologous UCB-derived product was transported to the operating room and thawed per protocol once the subject was off cardiopulmonary bypass and after anticoagulation reversal.

Manufacturing

Mononuclear cell enrichment was performed via Ficollbased density gradient purification using the Sepax 2 Cell Processing System (GE Healthcare). The Sepax 2 is a closed system for cell separation, washing and concentration. Following a final centrifugation step, the UCB-MNCS were suspended in CryoStor10, a commercially available dimethyl sulfoxide-based cryopreservation solution, at concentrations ranging from 12 to 42 million cells per mL. Sterile cryovials were manually filled after at processing. The number of vials in each batch is variable and depends on cell yield. Each vial is labelled with product specific identification and a unique vial specific identifier. After processing, each product undergoes product characterization testing to meet all pre-defined release criteria. Characterization and process markers captured are hours from collection to finish, CD45, CD34, cell count, MNC total, MNC%, TNC, and viability.

Storage and transportation

The cells were stored under Good Manufacturing Practice (GMP) standards at the manufacturing facility. The products were frozen using a controlled-rate freezing process and subsequently transferred to liquid nitrogen (vapor phase) for long-term storage. For transportation, the cells were shipped in a liquid nitrogen dry shipper to the administration hospital by sponsor staff at the time of cell delivery, with no product storage at the clinical site. At the time of cell delivery, the cells are thawed by hand and immediately loaded into the delivery device with no additional manipulation. To safeguard against contamination or degradation, product delivery begins within 30 min of thawing. The procedure was canceled if a backup vial was unavailable in the surgical suite.

Product dose and administration

The subjects in the treatment group received a target dose of 1–3 million total nucleated cells per kg of body weight of the autologous UCB-MNC product via intramyocardial injections. A 27-gauge needle attached to a 1 mL syringe was used to administer the product via direct subepicardial injections of 0.1 mL per 1 kg of body weight to reach the target dose. Injections were given slowly over at least 5 s, followed by a 20-second pause with the needle left in place to maximize cell retention within the myocardium. The injections were spaced approximately 1 cm apart across the free wall of the right ventricle. The injection sites were documented on a diagram to record the approximate location of the injection sites relative to the coronary vasculature and filed within each subject's study file.

Cell dosing is based on subject weight, and the volume for injection to achieve the desired dose is calculated at the time of cell delivery. The target dose range was established according to experience with our preclinical studies based on variability with cell recovery in the animal models, and was used as the acceptable range for our initial phase I and the current phase II study.

The rationale for intramyocardial delivery through direct subepicardial injections was chosen based on minimizing the risk of cell delivery for these single ventricle patients. Because these patients are required to undergo open chest cardiac surgery, the epicardial surface is routinely exposed, and intramyocardial delivery at the time of surgery as an "add-on" was determined to be less invasive and less risky compared to intracoronary infusion, which would require a separate catheter-based procedure and its associated ionizing radiation. Intramyocardial injections of UCB-MNC and subsequent follow-up were conducted in preclinical studies, which confirmed the feasibility, safety, and efficacy of cell delivery.

Data acquisition

After consent and before stage II palliation surgery, demographic, baseline clinical, and laboratory data were collected, as were baseline quantitative measurements of cardiac function data with transthoracic echocardiography (TTE). Operative and postoperative data were also collected, including physical examination, daily vitals, and telemetry monitoring to identify arrhythmias and other adverse events; likewise, at one, three, and twelve months after surgery, physical examination, blood workup, and TTE were performed. The Infant Toddler Quality of Life Questionnaire-97 (ITQOL-97) and the Ages and Stages Questionnaire-3 (ASQ-3) scores were also recorded at baseline and the twelve-month follow-up. The collected data were filed into applicable electronic case report forms (eCRFs) in the electronic data capture system.

Echocardiography data

To measure our coprimary endpoints, we planned for each of the ninety-five subjects to undergo a TTE at baseline, at discharge, and at three and twelve months after surgery. Right ventricular cardiac function was measured through fractional area change (FAC) and longitudinal and circumferential strain. This study used a blinded imaging core laboratory to evaluate the echocardiography images. The imaging core lab comprises Mayo Clinic staff not affiliated with this clinical trial.

Endpoints

Our coprimary efficacy endpoints assessed right ventricular cardiac function changes from baseline to three- and twelve-months following stage II palliation surgery. This assessment was conducted by measuring FAC and circumferential and longitudinal strain. The secondary endpoints assessed changes in biomarkers of cardiac injury (troponin-T, BNP) at three and six hours post surgery. Additionally, the safety endpoints included SAEs; changes in overall health, measured by vital signs (weight, heart rate, and oxygen saturation) at three and twelve months; and the cumulative incidence of hospitalization at one and three months. Finally, we measured developmental abilities at twelve months with the ITQOL-97 and ASQ-3 assessment tools following stage II palliation surgery.

Statistical analysis

Continuous data are presented as medians (IQRs), and categorical data are presented as counts (%). The Kruskal-Wallis or Wilcoxon rank sum test was used for between-group comparisons of continuous data. Categorical data were compared between groups via the chi-square test. Analysis of covariance (ANCOVA) models were used to test multiple outcomes between groups. For all ANCOVA models, we calculated the change from baseline, tested the computed change between groups, and adjusted for the baseline value of the dependent variable. Additional covariates were added in a multivariable sensitivity analysis to account for potential effect modifiers. The manuscript presents the marginal means of the ANCOVA models as LS means. Spearman correlation was used to assess the relationship between characterization and process markers and the primary outcomes. The primary and secondary endpoints were defined a priori, with additional comparisons added to better describe the sample, baseline status, and group differences in this work. An alpha of 0.05 was prespecified, and coprimary endpoints were adjusted post hoc to preserve alpha using the Bonferroni adjustment. No other adjustments were made for multiple testing outside the coprimary endpoints, and these results should be considered exploratory and not inferential. All analyses were performed via SAS Software (version 9.4), SAS Institute, Inc.

Results

Between June 2019 and September 2021, ninety-five subjects were enrolled. Fifty subjects were successfully enrolled in the treatment group, while forty-five were enrolled as controls. Our primary analysis included an echocardiographic evaluation of right ventricular cardiac function; unfortunately, owing to the lack of echocardiography data (n = 8), poor image quality (n = 3), subject withdrawal (n = 2), and cardiac transplant (n = 1), primary endpoint data could only be obtained for eighty-one subjects, forty in the treatment group and forty-one in the control group (Fig. 1).

Baseline

Demographics, vital signs, blood workup, and cardiac function data are summarized in Table 1. Both groups were predominantly composed of white males, with an even more pronounced representation of the treatment group, and had comparable anthropometric measurements and vital signs. The subjects in the treatment group were younger, with a median age of 4.7 months ([IQR, 4.5, 5.1] compared with 5.2 months [IQR, 4.4, 6.2] in the control group, (p=.048)). There were no significant differences between the groups regarding baseline cardiac function, cardiac biomarkers, or blood workup.

Stage I palliation surgery with and without a valved RV-PA shunt accounted for 78.0% (44.0% nonvalved, 34.0% valved) of stage I surgeries in the treatment group and 82.2% (71.1% nonvalved, 11.1% valved) in the control group (p =.049). For stage II palliation surgery, the most common surgery type was the Glenn procedure, accounting for 88.0% of the patients in the treatment group and 93.3% in the control group. The Hemi-Fontan procedure accounted for 12.0% and 6.7%, of the two groups, respectively. During this procedure, subjects treated with the UCB-MNC product had a shorter cardiopulmonary bypass time, with a median duration of 64.0 min ([IQR, 38.0, 91.0] compared with 83.0 min [IQR, 51.0, 113.5] in the control group, p=.024]. No complications were reported during surgery.

The baseline assessment of parental and legal guardians' perceptions of their children's overall health and developmental abilities with the ITQOL-97 and ASQ-3 assessment tools revealed better scores for the gross motor and personal social domains in the ASQ-3 tool in the subjects in the treatment group. In contrast, no differences were reported in overall health with the ITQOL-97 (See Supplementary Tables S1 and S2, Additional file 1).

Co-primary endpoints

Right ventricular cardiac function at three (short-term) and twelve months (long-term)

The ANCOVA model did not reveal any statistically significant difference in FAC or circumferential strain in the short or long term. However, longitudinal strain demonstrated an unfavorable change in the treatment group in the short term (p=.032, unadjusted; p=.19, adjusted). The LS mean estimate was 0.539 (95% CI -0.607–1.686) for the treatment group and -1.240 (95% CI -2.387 - 0.094) for the control group (Table 2). The difference in longitudinal strain was not observed in the long term (p =.254, unadjusted; p = 1.00, adjusted).

To account for potential effect modifiers, we adjusted the ANCOVA model for baseline stage I surgery type, tricuspid regurgitation, and cardiopulmonary bypass time (Table 3). This adjustment proved that subjects receiving the UCB-MNC product experienced an unfavorable change in longitudinal strain in the short term (p=.009). The LS mean estimate for the UCB-MNC group was 0.646 (95% CI -0.493–1.786), whereas the LS mean estimate was – 1.555 (95% CI -2.695 - -0.415) for the control group. Like the initial analysis, no differences were observed in the longitudinal strain at twelve months or in the FAC or circumferential strain at three and twelve months.

Secondary endpoints

Cardiac injury

Compared with those at baseline, troponin T levels were higher at three and six hours after surgery in the treatment group [(0.790 vs. 0.595, p = .037 at 3 h) and (0.700 vs. 0.539, p = .032 at 6 h)]. While the levels remained elevated in both groups at three months (0.417 vs. 0.425), there was no significant difference between them (p = .994). By twelve months, the troponin levels had decreased in both groups (-0.024 vs. -0.025, p = .147), as shown in Table 3.

BNP levels were predictably elevated before surgery and declined as anticipated during follow-ups at three and twelve months. However, no significant differences were found between the groups (See Supplementary Table S3, Additional file 1). Spearman correlation analysis did not reveal any association between characterization and process markers and the co-primary endpoints in the treated group.

Safety endpoints

Adverse events

The subjects in the treatment group presented a 20% greater cumulative incidence of at least one severe adverse event (SAE) at three months (58.0% vs. 37.8%, p=.048) of enrollment. The difference was not maintained at one year of enrollment (68.0% vs. 53.3%, p=.143)—Table 4. Among those with at least one SAE, the median number of SAEs observed within one and three months of enrollment was 2.0 for subjects in the treatment group and 1.0 for those in the control group. Through one year of enrollment, the median SAEs remained at 2.0 for the treatment group and increased to 1.5 for the control group (Table 5). Figure 2 illustrates the incidence of adverse events at one year of enrollment by system organ in each group.

 Table 1
 Demographics, baseline clinical characteristics, cardiac function, cardiac biomarkers, blood workup, and type of stage I and stage II palliation surgery, including postoperative troponins and blood pressure measurement

	Treatment group (n = 50)	Control group (n=45)	<i>P-</i> value
Age, months			0.048*
Median (Q1, Q3)	4.7 (4.5, 5.1)	5.2 (4.4, 6.2)	
Gender			0.019 [†]
Female	11 (22.0%)	20 (44.4%)	
Male	39 (78.0%)	25 (55.6%)	
Race			0.273 [†]
White	36 (72.0%)	29 (64.4%)	
Black/African American	3 (6.0%)	3 (6.7%)	
Asian	3 (6.0%)	1 (2.2%)	
More than one	3 (6.0%)	2 (4.4%)	
Unknown, none or no reported	5 (10.0%)	10 (22.2%)	
Weight, kg			0.538*
Median (Q1, Q3)	6.0 (5.6, 6.7)	6.1 (5.7, 6.6)	
Height, cm			0.552*
Median (Q1, Q3)	62.0 (60.0, 65.0)	61.0 (59.0, 64.0)	
Heart Rate/Pulse, bpm			0.232*
Median (Q1, Q3)	123.5 (115, 137)	122 (110, 132)	
SBP, mmHq	95 (74.0, 131.0)	94.5 (74.0, 173.0)	0.544*
Median (Q1, Q3)			
DBP , mmHg Median (Q1, Q3)	50.5 (28.0, 89.0)	53.0 (29.0, 95.0)	0.787*
SBP at discharge, mmHg Median (O1, O3)	96.5 (69.0, 138.0)	94.0 (69.0, 135.0)	0.069*
DBP at discharge, mmHg Median (Q1, Q3)	50.0 (37.0, 89.0)	52.0 (34.0, 81.0)	0.222*
SPO2 (%)			0.500*
SPU2 (%)	00 E (77 04)	90.0 (79. 94)	0.590
Condian (QT, QS)	00.3 (77, 04)	00.0 (70, 64)	
Cardiac function as assessed by echocardiograph	18	15	0.062*
FAC, % (I)	40	45	0.905
Median (QT, Q3)	41.7 (37.1, 44.9)	41.9 (38.1, 45.0)	0.460*
Circumferential Strain, % (n)	43	39	0.463
Median (QT, Q3)	-15.3 (-17.7, -11.9)	-16.4 (-18.9, -12.5)	0.127*
Longitudinal Strain, % (n)	48	45	0.137
Median (Q1, Q3)	-14.5 (-17.0, -12.4)	-15.9 (-18.0, -13.4)	
Cardiac biomarkers	10		o o = c*
Iroponin-I, ng/mL (n)	42	41	0.376
Median (Q1, Q3)	0.025 (0.018, 0.039)	0.024 (0.013, 0.038)	0.404*
Iroponin-I, 3 h post-surgery, ng/mL (n) Median (Q1, Q3)	42 0.797 (0.136, 2.322)	40 0.643 (0.155, 1.424)	0.131
Troponin-T , 6 h post-surgery , ng/mL (n) Median (Q1, Q3)	46 0.666 (0.000, 1.763)	39 0.583 (0.015)	0.255*
NT-Pro-BNP, pg/mL (n)	48	43	0.450*
Median (Q1, Q3)	1296.0 (687.5, 3066.0)	1200.0 (855.0, 2196.0)	
CRP, mg/L (n)	49	42	0.105*
Median (Q1, Q3)	5.0 (2.9, 5.0)	5.0 (2.5, 6.0)	
Liver/Renal Function			
ALT, U/L (n)	48	42	0.420*
Median (Q1, Q3)	25.0 (19.1, 34.0)	22.0 (17.0, 37.0)	
AST, U/L (n)	48	42	0.938*
Median (Q1, Q3)	43.0 (36.0, 53.0)	40.5 (36.0, 56.0)	
Alk Phosphatase, U/L (n)	48	42	0.087*
Median (Q1, Q3)	269.5 (222.0, 333.5)	237.0 (197.0, 309.0)	

Table 1 (continued)

	Treatment group	Control group	P-
	(<i>n</i> =50)	(n=45)	value
Bilirubin, mg/dL (n)	48	42	0.948*
Median (Q1, Q3)	0.4 (0.3, 0.5)	0.4 (0.3, 0.6)	
Albumin, g/dL (n)	48	43	0.529*
Median (Q1, Q3)	3.5 (3.2, 4.2)	3.5 (3.1, 4.1)	
Total proteins, g/dL (n)	48	41	0.211*
Median (Q1, Q3)	5.6 (5.3, 6.1)	5.4 (5.0, 5.9)	
Creatinine, mg/dL (n)	48	43	0.049*
Median (Q1, Q3)	0.2 (0.2, 0.3)	0.2 (0.2, 0.2)	
BUN, mg/dL (n)	48	43	0.503*
Median (Q1, Q3)	11.0 (8.0, 15.0)	12.0 (8.0, 16.0)	
CBC w/Differential			
Hemoglobin, g/dL (n)	49	45	0.365*
Median (Q1, Q3)	15.1 (13.9, 16.1)	15.6 (14.2, 16.3)	
Hematocrit, % (n)	49	45	0.386*
Median (Q1, Q3)	46.1 (42.9, 50.0)	46.6 (43.9, 50.6)	
WBCs, (n)	49	45	0.671*
Median (Q1, Q3)	7.2 (5.7, 8.6)	7.2 (5.8, 8.9)	
Platelets, (n)	49	45	0.379*
Median (Q1, Q3)	332.0 (278.0, 390.0)	303.0 (264.0, 360.0)	
Lymphocytes, (n)	47	45	0.046*
Median (Q1, Q3)	3.5 (2.4, 5.6)	3.1 (2.1, 4.1)	
Type of stage I and stage II palliation surgery			
Stage I surgery type			0.049 [†]
Norwood procedure with mBT shunt	8 (16.0%)	5 (11.1%)	
Norwood procedure	22 (44 0%)	32 (71 1%)	
with nonvalved RV- PA shunt	(```````		
Norwood procedure with	17 (34.0%)	5 (11.1%)	
Valved RV-PA Shufit	2 (4 00()	1 (2 20/)	
Other (creatify)	2 (4.0%)	1 (2.2%)	
	1 (2.0%)	2 (4.4%)	0.275
Glapp procedure	44 (88 004)	42 (02 20%)	0.375
Hemi Fontan	44 (08.0%) 6 (12.0%)	42 (93.370)	
Stage II Surgery Details	0 (12.070)	5 (0.7%)	
Stage II Surgery Details	40	4.4	0.024*
time min (n)	40	44	0.024
Median (01, 03)	64.0 (38.0, 91.0)	83.0 (51.0. 113.5)	
Aortic cross-clamp	19	22	0.732*
time, min (n)			0 52
Median (Q1, Q3)	28.0 (0.0, 50.0)	3.0 (0.0, 53.0)	

CRP: C-Reactive protein; DBP: Diastolic blood pressure, mBT shunt: modified Blalock-Taussig shunt, RV-PA shunt: Right ventricular pulmonary artery shunt, SBP: Systolic blood pressure, WBC: White blood cell count

*Kruskal–Wallis

[†]Chi Square

Changes in overall health and cumulative hospitalization

No statistically significant differences were observed in any of the vital signs (weight, heart rate, or oxygen saturation) used to assess changes in overall health, either in the short or long term. Overall, patients experienced an increase in weight of 1.2 kg at three months and 3.5 kg at twelve months. Heart rate and oxygen saturation did not significantly change in any of the groups. The detailed results of these measurements are presented in Table 6. Additionally, the two groups had no significant difference in cumulative hospitalization. The median number of days in the hospital was 8.5 (IQR 6.0, 22.0) for the treatment group and 10.0 (IQR 6.0, 16.0) days for the control group, as shown in Table 6.

monuns						
Visits Compared	Cardiac Function	n	Treatment group LS Means Estimate (95% CI)	n	Control group LS Means Estimate (95% Cl)	P value (adjusted [*])
Preop to 3 months	Apical FAC	40	-2.225 (-3.898, -0.552)	41	-2.511 (-4.164, -0.895)	0.80 (1.00)
	Circumferen tial Strain	30	1.341 (-0.310, 2.992)	35	0.170 (-1.358, 1.699)	0.302 (1.00)
	Longitudinal Strain	39	0.539 (-0.607, 1.686)	39	-1.240 (-2.387, -0.094)	0.032 (0.19)
Preop to 12 months	Apical FAC	33	-3.897 (-5.680, -2.114)	32	-4.431 (-6.241, -2.620)	0.676 (1.00)
	Circumferen tial Strain	24	0.852 -1.001, 2.706)	20	2.159 (0.128, 4.190)	0.343 (1.00)
	Longitudinal Strain	31	0.516 (-0.638, 1.715)	32	-0.453 (-1.633, 0.727)	0.254 (1.00)

Table 2 ANCOVA model for primary endpoints. Right ventricular cardiac function was adjusted for baseline values at three and twelve months

ANCOVA: Analysis of covariance; FAC: Fractional area change

*Primary endpoints adjusted via the Bonferroni method to preserve the type I error rate

Table 3 ANCOVA model for primary endpoints and troponins adjusted for baseline tricuspid regurgitation, stage I surgical palliation, and cardiopulmonary bypass time

Visits Compared	Cardiac Function	n	Treatment group LS Means Estimate (95% CI)	n	Control group LS Means Estimate (95% Cl)	<i>P</i> value
Preop to 3 h postsurgery	Troponin T	36	0.790 (0.665, 0.915)	36	0.595 (0.470, 0.720)	0.037
Preop to 6 h postsurgery	Troponin T	39	0.700 (0.602, 0.798)	35	0.539 (0.436, 0.643)	0.032
Preop to 3 months	Apical FAC	39	-2.401 (-4.144, -0.658)	40	-2.444 (-4.164, -0.723)	0.972
	Circumferential Strain	30	1.216 (-0.489, 2.922)	34	0.070 (-1.526, 1.667)	0.342
	Longitudinal Strain	38	0.646 (-0.493, 1.786)	38	-1.555 (-2.695, -0.415)	0.009
	Troponin T	13	0.417 (-1.154, 1.988)	14	0.425 (-1.075, 1.926)	0.994
Preop to 12 months	Apical FAC	33	-3.792 (-5.589, -1.994)	31	-4.598 (-6.454, -2.743)	0.536
	Circumferential Strain	24	0.957 (-1.037, 2.953)	19	1.887 (-0.372, 4.147)	0.548
	Longitudinal Strain	31	0.489 (-0.737, 1.715)	31	-0.495 (-1.722, 0.731)	0.263
	Troponin T	13	-0.024 (-0.025, -0.023)	11	-0.025 (-0.026, -0.024)	0.147

ANCOVA: analysis of covariance; FAC: fractional area change

ITQOL-97 and ASQ-3 assessment tools

Compared with the baseline ASQ-3 assessment tool, which revealed greater performance in the gross motor and personal social domains for the UCB-MNC group, no statistically significant differences were noted in these domains or any of the remaining domains assessed by the ASQ-3 and the 13 domains evaluated by the ITQOL-97 assessment tool at twelve months. (See Supplementary Tables S4 and S5, Additional file 1).

Fatalities

At the one-year follow-up, five subjects had died, three in the treatment group and two in the control group; none prompted the activation of any adverse eventstopping rule, and none triggered mandatory reporting to the FDA. The causes of death involved multisystem failure, bradycardic arrest, and respiratory and heart failure. Notably, all these incidents occurred more than 60 days after surgery, ranging from five to twelve months. Notably, the site principal investigator deemed all deaths unrelated to the product or delivery procedure. The Data

 Table 4
 Serious adverse events, comparison by group

	Treatment group (n=50)	Control group (n=45)	P value
Did the subject have at least one SAE within 1- month of enrollment?			0.0568*
Yes	24 (48.0%)	13 (28.9%)	
No	26 (52.0%)	32 (71.1%)	
Did the subject have at least one SAE within 3- months of enrollment?			0.048*
Yes	29 (58.0%)	17 (37.8%)	
No	21 (42.0%)	28 (62.2%)	
Did the subject have at least one SAE within 12- months of enrollment?			0.143*
Yes	34 (68.0%)	24 (53.3%)	
No	16 (32.0%)	21 (46.7%)	
SAE: Serious adverse event *Chi-Square p value.			

Safety and Monitoring Board did not recommend stopping or modifying the trial because of safety concerns.

Discussion

This phase IIb clinical trial was designed as a follow-up study to support and broaden the findings of our previous phase I trial, which was published in 2019 and later expanded in 2021. This trial examined the safety and feasibility of intramyocardial injections of an autologous UCB-MNC product in HLHS patients [18, 19]. This initial study revealed no safety concerns over the six-month follow-up, along with preservation of right ventricular cardiac function and normalized growth rates in ten patients. In the present study, we further wanted to evaluate the efficacy of the autologous UCB-MNC product in a controlled trial. Our findings did not reveal significantly superior cardiac outcomes determined by detrimental cardiac strain in the shortterm following product administration. While safety endpoints were evaluated, our assessment was not exhaustive. Notably, our study uncovered no safety issues or concerns associated with the product's usage. Additionally, we did not observe any We generated the first data-based efficacy clinical study concerning autologous intramyocardial injections of a UCB-MNC product in patients with HLHS. This study is also the largest reported clinical trial of stem cell therapy in HLHS patients.

Although it is an emerging field, significant advances and ongoing efforts have encouraged our urge to provide more robust evidence and strategies to help guide clinical decisions. In 2017, the PERSEUS clinical trial, a phase IIb study, included thirty-four subjects receiving intracoronary infusion of autologous cardiac progenitor cells [20]. They concluded that there was a significant and favorable improvement in cardiac function among the treated individuals, as observed through various imaging modalities at three and twelve months. This beneficial effect was also noticeable in late-treated subjects. Notably, at twelve months, all groups demonstrated an increase in the ejection fraction, as evidenced by increased global strain and strain rate and reduced ventricular volume. An additional study, the ELPIS phase I clinical trial, probed safety and suggested favorable right ventricular performance at one year of intramyocardial cell-based therapy using allogeneic bone marrow-derived mesenchymal stem cells (Lomecel-B) during stage II palliation surgery. As phase IIb trials continue to enroll patients, the medical community is encouraged by the potential of these findings.

In recent years, extensive evidence has supported stem cell therapies from perinatal sources, such as UCB and umbilical cord tissue (UCT). Products derived from these sources have shown a positive impact on the cardiovas-cular system, primarily through regenerative paracrine mechanisms [21–23]. Our study focused on developing a product derived from UCB-MNCs, combining various stem cell types to enhance therapeutic outcomes.

While UCB primarily contains hematopoietic cells, it also contains mesenchymal stem cells (MSCs), which is not a primary source. Historical evidence indicates that UCB is a challenging source of MSCs due to the low cell

 Table 5
 Serious adverse events, comparison by groups among those with one or more SAEs

	Treatment group (n = 50)	Control group (n=45)	P value
Number of SAEs within 1- month of enrollment (n)	24	13	0.17*
Median (Q1, Q3)	2.0 (1.0, 6.0)	1.0 (1.0, 4.0)	
Number of SAEs within 3- months of enrollment (n)	29	17	0.18*
Median (Q1, Q3)	2.0 (1.0, 8.0)	1.0 (1.0, 7.0)	
Number of SAEs within 12- months of enrollment (n)	34	24	0.16*
Median (Q1, Q3)	2.0 (1.0, 13.0)	1.5 (1.0, 9.0)	
SAEs: Serious adverse events *Wilcoxon rank sum p value.			



Fig. 2 Adverse event incidence within one year of enrollment by system organ

Table 6 Cumulative hospitalization at one and three months and changes in vital signs from baseline to three and twelve months after surgery

	Treatment group	Control group	P value
	(n=50)	(<i>n</i> =45)	
Change in vital signs (weight, heart rate	e, O2 saturation)		
Change in weight at	44	39	0.956*
3 months (n)			
Median (Q1, Q3)	1.2 (0.9, 1.5)	1.2 (0.7, 1.8)	
Change in weight at	40	36	0.277*
12 months (n)			
Median (Q1, Q3)	3.5 (2.9, 4.3)	3.5 (2.3, 4.2)	
Change in heart rate at	43	38	0.909*
3 months (n)			
Median (Q1, Q3)	5.0 (-8.0, 14.0)	3.0 (-9.0, 17.0)	
Change in heart rate at	37	34	0.259*
12 months (n)			
Median (Q1, Q3)	-10.0 (-13.0, 8.0)	1.0 (-12.0, 14.0)	
Change in O2 saturation at	40	36	0.639*
3 months (n)			
Median (Q1, Q3)	4.0 (1.0, 8.5)	5.5 (0.5, 9.5)	
Change in O2 saturation at	38	29	0.301*
12 months (n)			
Median (Q1, Q3)	4.0 (0.0. 9.0)	2.0 (-3.0, 8.0)	
Cumulative Hospitalization			
At 1 month, days			0.776*
Median (Q1, Q3)	8.5 (6.0, 22.0)	10.0 (6.0, 16.0)	
At 3 months, days			0.878*
Median (Q1, Q3)	8.5 (6.0, 22.0)	10.0 (6.0, 16.0)	
*Kruskal–Wallis.			

counts per cord blood unit and their limited proliferation rates compared to cord tissue [24–26]. In contrast, UCT is rich in MSCs, which can differentiate into various cell types, including adipocytes, osteoblasts, myocytes, and others. These limitations pose a significant barrier to utilizing human MSCs from UCB in regenerative medicine. Furthermore, MSCs derived from UCT differ significantly in functional properties and characterization compared to those derived from UCB. Products from UCT require additional processing steps such as extraction and cellular expansion to achieve therapeutic outcomes.

MSCs from UCT have shown the ability to produce exosomes, which is an acellular product that according to the evidence, is responsible for triggering molecular cascades involved in muscle repair, either cardiac or skeletal [27–29]. Both UCB-MNCs and MSCs from UCT share a promising potential for cardiovascular regeneration, with each therapy offering unique regenerative and anti-inflammatory benefits [30]. Together, they represent complementary approaches that aim to repair damaged cardiac tissues, enhance healing, and ultimately improve patient outcomes, paving the way for a future where regenerative medicine becomes a standard in cardiovascular care.

Our study has several notable strengths. Its multicenter design from multiple dispersed pediatric hospitals around the United States facilitated the generation of a substantial sample size, thereby enhancing the applicability and generalizability of the findings to real-world scenarios.

Our results did not provide evidence to demonstrate advantages in cardiac function and support our hypothesis explicitly. Instead, we observed that the administration of the autologous UCB-MNC product seemed to detrimentally affect right ventricular cardiac strain in the short term compared with the controls. Despite the shorter cardiopulmonary bypass (CPB) time in the treatment group, which could raise concerns regarding its association with the administration of the product, the analysis was adjusted for CPB time, among other covariates, to mitigate any confounding effects. As a result, the initially observed detrimental cardiac function persisted only in the short term. We hypothesize that this detrimental effect could be related to transient and localized needle injury during product administration, which is subclinical and not damaging enough to persist in time and increase the level of cardiac biomarkers. This apparent cardiac dysfunction was resolved by twelve-month echocardiography.

Furthermore, the UCB collection protocol of this study is aligned with that of our phase I clinical trial, ensuring a collection of over 35 mL of UCB that guarantees a success rate exceeding 90%. Despite previous assertions that this study represents the largest clinical trial on stem cell therapy for HLHS, there might be an essential sample size factor limiting the outcomes. In 2020, the FUEL trial in a cohort of 400 subjects with single-ventricle CHD who had undergone Fontan surgery and were treated with either udenafil or placebo at 26 weeks did not detect a difference between groups when the myocardial performance index was assessed [31]. These findings might suggest that, despite a large sample size, additional measures might be necessary to identify subtle differences. We recommend an extended follow-up, which could yield long-term benefits beyond the current observations and contribute encouraging evidence to the literature on stem cell therapy.

Additionally, we observed a decrease of up to 6% in the apical FAC in the control group at twelve months, which is consistent with our previous findings using natural history data, which reported a 5% decrease in FAC six months after stage II palliation surgery in controls [19]. While cMRI remains the gold standard for evaluating right ventricular cardiac function in CHD patients because of its superior spatial visualization of anatomical structures, studies support the complementary roles of 2-D echocardiography and cMRI, each contributing to limitations [32, 33]. Our data support the natural history of right ventricular cardiac function via echocardiography and suggest more assertive imaging methods to further evaluate FAC and other measures of RV cardiac function, volume, and size.

Limitations

First, the primary analysis relied on the standard of care echocardiography as the imaging assessment tool. While this method is dependable, it may provide less precise and accurate right ventricular cardiac function measurements than a more thorough and reproducible tool such as cardiac magnetic resonance imaging (cMRI). Although cMRI was specified, we amended our protocol during the COVID-19 pandemic to make cMRI optional at followup to comply with restrictions on elective procedures at most clinical centers. We do not have sufficient cMRI to perform a meaningful statistical assessment. Second, the study design focused exclusively on analyzing a homogenous cohort of patients with preserved baseline cardiac function, excluding those with abnormal baseline right ventricular cardiac function, which would have allowed for a more sensitive assessment of the potential benefit of the UCB-MNC product. Third, this study was nonrandomized to best utilize valuable UCB-MNC products, introducing inherent biases that we tried to minimize through rigorous patient selection and a blinded assessment of imaging outcomes, ensuring the reliability of our findings.

Our continuous efforts provide fertile ground for further investigation, encouraging researchers to identify areas for improvement, assess strengths and weaknesses, and explore potential strategies. This endeavor aims to facilitate reevaluating existing approaches and developing novel hypotheses, shedding light on the practical applicability of UCB-MNC products.

Conclusions

In conclusion, this phase IIb clinical trial of intramyocardial injections of autologous UCB-MNC products did not reveal superior cardiac outcomes in HLHS patients during stage II palliation surgery. There was an unfavorable outcome regarding cardiac function, as the product failed to improve right ventricular cardiac strain in the short and long term compared with controls. The safety profile of the UCB-MNC product remains acceptable, and no serious safety issues or concerns have been identified. Further randomized, placebo-controlled, and blinded studies addressing our limitations are needed. Our team is committed to continuing to study the effectiveness of novel stem cell therapies, specifically UCB-MNC, in children with CHD. We are undertaking novel initiatives to overcome study limitations and facilitate a more precise and comprehensive evaluation of the UCB-MNC product's long-term benefits. Additionally, we aimed to assess other critical long-term outcomes, such as completing stage III palliation and the time to heart transplantation.

Abbreviations

ANCOVA	Analysis of covariance
ASQ-3	Age and Stage Questionnaire-3
cMRI	Cardiac magnetic resonance imaging
DMSO	Dimethyl sulfoxide
eCRF	Electronic Case Report Form
FAC	Fractional area change
HLHS	Hypoplastic left heart syndrome
IRB	Institutional Review Board
ITQOL-97	Infant Toddler Quality of Life Questionnaire-97
mBT shunt	Modified Blalock-Taussig shunt
RV-PA shunt	Right ventricular pulmonary artery shunt
SAEs	Severe adverse events
UCB-MNC	Umbilical Cord Blood-derived Mononuclear Cells

Supplementary Information

The online version contains supplementary material available at https://doi.or g/10.1186/s13287-025-04316-3.

```
Supplementary Material 1
```

Acknowledgements

We want to express our gratitude to the entire team of people within our HLHS consortium for their invaluable contributions to data collection, the manufacturing process, administrative support, surgical assistance, and technicians, among others, which were crucial for the realization of this clinical trial.

Author contributions

Design of the study (TJN, SCP, CH), data collection (HLHS consortium), statistical analysis (DWS), writing-original draft and elaboration of tables (CGN, CH, DWS), writing-review & editing (TJN, CH), critical review (HMB, WFC, DLC, MYQ, JWR). All authors read and approved the last version of the manuscript and agreed to be responsible for all aspects of the clinical trial.

Funding

This study was supported by the Todd and Karen Wanek Family Program for HLHS at the Mayo Clinic.

Data availability

The datasets generated and analyzed during the current study are available from the corresponding author on reasonable request. All materials used in the study are available upon request.

Declarations

Ethics approval and consent to participate

The study titled "Phase IIb Study of Intramyocardial Injection of Autologous Umbilical Cord Blood-Derived Mononuclear Cells during Stage II Surgical Repair of Hypoplastic Left Heart Syndrome (Auto Cell-II)" was individually approved by the Institutional Review Boards (IRBs) at each participating site under protocol number CSP-4401 and IND 15343 on August 30, 2018.

Consent for publication

Not applicable.

Competing interests

The authors declare that they have no competing interests.

Artificial intelligence

The authors declare that they have not used Al-generated work in this manuscript.

Author details

¹Division of Cardiovascular Diseases, Center for Regenerative Medicine, Mayo Clinic, Rochester, MN, USA

 $^{\rm 2}{\rm Program}$ for Hypoplastic Left Heart Syndrome, Mayo Clinic Rochester, Rochester, MN, USA

³Division of Congenital Heart Surgery, Heart Institute, Children's Hospital Colorado, University of Colorado Denver Anschutz Medical Campus, Denver, CO, USA

⁴Division of Cardiac, Thoracic and Vascular Surgery, University of Oklahoma Health Sciences, Oklahoma, USA

⁵Division of Pediatric Cardiology, University of Alabama Birmingham, Birmingham, AL, USA

⁶Division of Congenital Heart Surgery, Heart Institute, Cincinnati Children's Hospital Medical Center, Cincinnati, OH, USA

⁷Division of Pediatric Cardiology, Department of Pediatric and Adolescent Medicine, Mayo Clinic, Rochester, MN, USA

⁸Department of Pediatrics, Division of Cardiology, Children's Hospital of Philadelphia, Philadelphia, USA

⁹HeartWorks Inc. Rochester, Rochester, MN, USA

¹⁰General Internal Medicine, Department of Molecular Pharmacology and Experimental Therapeutics, Center for Regenerative Medicine, Mayo Clinic, Rochester, MN, USA

Received: 25 November 2024 / Accepted: 7 April 2025 Published online: 01 May 2025

References

- van der Linde D, Konings EEM, Slager MA, Witsenburg M, Helbing WA, Takkenberg JJM et al. Birth prevalence of congenital heart disease worldwide: a systematic review and meta-analysis. J Am Coll Cardiol [Internet]. 2011;58(21):2241–7. Available from: https://doi.org/10.1016/j.jacc.2011.08.025
- Virani SS, Alonso A, Aparicio HJ, Benjamin EJ, Bittencourt MS, Callaway CW et al. Heart Disease and Stroke Statistics-2021 Update: A Report From the American Heart Association. Circulation [Internet]. 2021;143(8):e254–743. Available from: https://doi.org/10.1161/CIR.00000000000950
- De Rita F, Crossland D, Griselli M, Hasan A. Management of the failing Fontan. Semin Thorac Cardiovasc Surg Pediatr Card Surg Annu [Internet]. 2015;18(1):2–6. Available from: https://doi.org/10.1053/j.pcsu.2015.01.004

- Gordon BM, Rodriguez S, Lee M, Chang R-K. Decreasing number of deaths of infants with hypoplastic left heart syndrome. J Pediatr [Internet]. 2008;153(3):354–8. Available from: https://doi.org/10.1016/j.jpeds.2008.03.00
- Barron DJ, Kilby MD, Davies B, Wright JGC, Jones TJ, Brawn WJ. Hypoplastic left heart syndrome. Lancet [Internet]. 2009;374(9689):551–64. Available from: htt ps://doi.org/10.1016/S0140-6736(09)60563-8
- Sano S, Ishino K, Kawada M, Arai S, Kasahara S, Asai T et al. Right ventriclepulmonary artery shunt in first-stage palliation of hypoplastic left heart syndrome. J Thorac Cardiovasc Surg [Internet]. 2003;126(2):504–9; discussion 509–10. Available from: https://doi.org/10.1016/s0022-5223(02)73575-7
- Ohye RG, Schranz D, D'Udekem Y. Current Therapy for Hypoplastic Left Heart Syndrome and Related Single Ventricle Lesions. Circulation [Internet]. 2016;134(17):1265–79. Available from: https://doi.org/10.1161/CIRCULATION AHA.116.022816
- Metcalf MK, Rychik J. Outcomes in Hypoplastic Left Heart Syndrome. Pediatr Clin North Am [Internet]. 2020;67(5):945–62. Available from: https://doi.org/1 0.1016/j.pcl.2020.06.008
- Rychik J, Goldberg DJ. Late consequences of the Fontan operation. Circulation [Internet]. 2014;130(17):1525–8. Available from: https://doi.org/10.1161/ CIRCULATIONAHA.114.005341
- Gewillig M, Brown SC, van de Bruaene A, Rychik J. Providing a framework of principles for conceptualising the Fontan circulation. Acta Paediatr [Internet]. 2020;109(4):651–8. Available from: https://doi.org/10.1111/apa.15098
- Bittle GJ, Morales D, Deatrick KB, Parchment N, Saha P, Mishra R et al. Stem Cell Therapy for Hypoplastic Left Heart Syndrome: Mechanism, Clinical Application, and Future Directions. Circ Res [Internet]. 2018;123(2):288–300. Available from: https://doi.org/10.1161/CIRCRESAHA.117.311206
- Ishigami S, Ohtsuki S, Tarui S, Ousaka D, Eitoku T, Kondo M et al. Intracoronary autologous cardiac progenitor cell transfer in patients with hypoplastic left heart syndrome: the TICAP prospective phase 1 controlled trial. Circ Res [Internet]. 2015;116(4):653–64. Available from: https://doi.org/10.1161/CIRCR ESAHA.116.304671
- Kaushal S, Wehman B, Pietris N, Naughton C, Bentzen SM, Bigham G et al. Study design and rationale for ELPIS: A phase I/IIb randomized pilot study of allogeneic human mesenchymal stem cell injection in patients with hypoplastic left heart syndrome. Am Heart J [Internet]. 2017;192:48–56. Available from: https://www.sciencedirect.com/science/article/pii/S000287031730182
- 14. de Jong R, Houtgraaf JH, Samiei S, Boersma E, Duckers HJ. Intracoronary stem cell infusion after acute myocardial infarction: a meta-analysis and update on clinical trials. Circ Cardiovasc Interv [Internet]. 2014;7(2):156–67. Available from: https://doi.org/10.1161/CIRCINTERVENTIONS.113.001009
- Delewi R, Hirsch A, Tijssen JG, Schächinger V, Wojakowski W, Roncalli J et al. Impact of intracoronary bone marrow cell therapy on left ventricular function in the setting of ST-segment elevation myocardial infarction: a collaborative meta-analysis. Eur Heart J [Internet]. 2014;35(15):989–98. Available from: https ://doi.org/10.1093/eurheartj/eht372
- Zimmet H, Porapakkham P, Porapakkham P, Sata Y, Haas SJ, Itescu S et al. Short- and long-term outcomes of intracoronary and endogenously mobilized bone marrow stem cells in the treatment of ST-segment elevation myocardial infarction: a meta-analysis of randomized control trials. Eur J Heart Fail [Internet]. 2012;14(1):91–105. Available from: https://doi.org/10.1093/eurj hf/hfr148
- Makkar RR, Smith RR, Cheng K, Malliaras K, Thomson LE, Berman D et al. Intracoronary cardiosphere-derived cells for heart regeneration after myocardial infarction (CADUCEUS): a prospective, randomised phase 1 trial. Lancet [Internet]. 2012;379(9819):895–904. Available from: https://doi.org/10.1016/S 0140-6736(12)60195-0
- Burkhart HM, Qureshi MY, Rossano JW, Cantero Peral S, O'Leary PW, Hathcock M et al. Autologous stem cell therapy for hypoplastic left heart syndrome: Safety and feasibility of intraoperative intramyocardial injections. J Thorac Cardiovasc Surg [Internet]. 2019;158(6):1614–23. Available from: https://doi.or g/10.1016/j.jtcvs.2019.06.001
- Vincenti M, O'Leary PW, Qureshi MY, Seisler DK, Burkhart HM, Cetta F et al. Clinical Impact of Autologous Cell Therapy on Hypoplastic Left Heart Syndrome After Bidirectional Cavopulmonary Anastomosis. Semin Thorac Cardiovasc Surg [Internet]. 2021 Autumn;33(3):791–801. Available from: https: //doi.org/10.1053/j.semtcvs.2020.11.002

- Ishigami S, Ohtsuki S, Eitoku T, Ousaka D, Kondo M, Kurita Y et al. Intracoronary Cardiac Progenitor Cells in Single Ventricle Physiology: The PERSEUS (Cardiac Progenitor Cell Infusion to Treat Univentricular Heart Disease) Randomized Phase 2 Trial. Circ Res [Internet]. 2017;120(7):1162–73. Available from: https://doi.org/10.1161/CIRCRESAHA.116.310253
- Santos Nascimento D, Mosqueira D, Sousa LM, Teixeira M, Filipe M, Resende TP et al. Human umbilical cord tissue-derived mesenchymal stromal cells attenuate remodeling after myocardial infarction by proangiogenic, antiapoptotic, and endogenous cell-activation mechanisms. Stem Cell Res Ther [Internet]. 2014;5(1):5. Available from: https://doi.org/10.1186/scrt394
- López Y, Lutjemeier B, Seshareddy K, Trevino EM, Hageman KS, Musch TI et al. Wharton's jelly or bone marrow mesenchymal stromal cells improve cardiac function following myocardial infarction for more than 32 weeks in a rat model: a preliminary report. Curr Stem Cell Res Ther [Internet]. 2013;8(1):46– 59. Available from: https://doi.org/10.2174/1574888x11308010007
- Balbi C, Bollini S. Fetal and perinatal stem cells in cardiac regeneration: Moving forward to the paracrine era. Placenta [Internet]. 2017;59:96–106. Available from: https://doi.org/10.1016/j.placenta.2017.04.008
- Mareschi K, Biasin E, Piacibello W, Aglietta M, Madon E, Fagioli F. Isolation of human mesenchymal stem cells: bone marrow versus umbilical cord blood. Haematologica [Internet]. 2001;86(10):1099–100. Available from: https://www .ncbi.nlm.nih.gov/pubmed/11602418
- Musina RA, Bekchanova ES, Belyavskii AV, Grinenko TS, Sukhikh GT. Umbilical cord blood mesenchymal stem cells. Bull Exp Biol Med [Internet]. 2007;143(1):127–31. Available from: https://doi.org/10.1007/s10517-007-003 2-z
- Fong C-Y, Subramanian A, Biswas A, Gauthaman K, Srikanth P, Hande MP et al. Derivation efficiency, cell proliferation, freeze-thaw survival, stem-cell properties and differentiation of human Wharton's jelly stem cells. Reprod Biomed Online [Internet]. 2010;21(3):391–401. Available from: https://doi.org/10.1016 /j.rbmo.2010.04.010
- Wu Y, Li J, Feng K, Tan A, Gao Y, Chen W et al. N-CADHERIN+/CD168- subpopulation determines therapeutic variations of UC-MSCs for cardiac repair after myocardial infarction. Stem Cell Res Ther [Internet]. 2024;15(1):423. Available from: https://doi.org/10.1186/s13287-024-04032-4
- Zhao L, Liu X, Zhang Y, Liang X, Ding Y, Xu Y et al. Enhanced cell survival and paracrine effects of mesenchymal stem cells overexpressing hepatocyte growth factor promote cardioprotection in myocardial infarction. Exp Cell Res [Internet]. 2016;344(1):30–9. Available from: https://doi.org/10.1016/j.yexc r.2016.03.024
- Piao L, Huang Z, Inoue A, Kuzuya M, Cheng XW. Human umbilical cordderived mesenchymal stromal cells ameliorate aging-associated skeletal muscle atrophy and dysfunction by modulating apoptosis and mitochondrial damage in SAMP10 mice. Stem Cell Res Ther [Internet]. 2022;13(1):226. Available from: https://doi.org/10.1186/s13287-022-02895-z
- Oommen S, Yamada S, Cantero Peral S, Campbell KA, Bruinsma ES, Terzic A et al. Human umbilical cord blood-derived mononuclear cells improve murine ventricular function upon intramyocardial delivery in right ventricular chronic pressure overload. Stem Cell Res Ther [Internet]. 2015;6(1). Available from: htt ps://doi.org/10.1186/s13287-015-0044-y
- Di Maria MV, Goldberg DJ, Zak V, Hu C, Lubert AM, Dragulescu A et al. Impact of Udenafil on Echocardiographic Indices of Single Ventricle Size and Function in FUEL Study Participants. Circ Cardiovasc Imaging [Internet]. 2022;15(11):e013676. Available from: https://doi.org/10.1161/CIRCIMAGING.1 21.013676
- Yeh DD, Foster E. Is MRI the Preferred Method for Evaluating Right Ventricular Size and Function in Patients With Congenital Heart Disease? Circ Cardiovasc Imaging [Internet]. 2014;7(1):198–205. Available from: https://doi.org/10.1161 /CIRCIMAGING.113.000395
- 33. Geva T. Is MRI the preferred method for evaluating right ventricular size and function in patients with congenital heart disease? MRI is the preferred method for evaluating right ventricular size and function in patients with congenital heart disease. Circ Cardiovasc Imaging [Internet]. 2014;7(1):190–7. Available from: https://doi.org/10.1161/CIRCIMAGING.113.000553

Publisher's note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.